Author’s response to reviews

Title: A Treatment Applying a Biomechanical Device to the Feet of Patients with Knee OA Results in Reduced Pain and Improved Function. A Prospective Controlled Study

Authors:

Yaron Bar-Ziv (ybarziv@013.net.il)
Yiftah Beer (yiftah.beer@gmail.com)
Yuval Ran (debyroay@012.net.il)
Shaike Benedict (Shaike12@walla.com)
Nahum Halperin (nhalperin@post.tau.ac.il)

Version: 3 Date: 20 June 2010

Author’s response to reviews: see over
Dear Dr. Shipley,

These are the revisions to our work: "A Treatment Applying a Biomechanical Device to the Feet of Patients with Knee OA Results in Reduced Pain and Improved Function. A Prospective Controlled Study" (previously named: “A novel device and treatment methodology for reduced pain and improved function in knee osteoarthritis: a prospective controlled study.”)

We have clarified the committee for ethical approval of the study and added the NIH clinical trial registration number to the manuscript.

Page 5, line 16: The protocol was approved by the Institutional Helsinki Committee Registry of Assaf Harofe Medical Center (Helsinki registration number 44/05 and NIH clinical trial registration number NCT00457132).

Sincerely,

Dr. Yuval Ran (corresponding author)
E-mail: debyroay@012.net.il